PERFORMANCE EVALUATION OF AGRIBAZAAR

FAUDZIAH AHMAD WAN ROZAINI SHEIK OSMAN NOR FARZANA ABD GHANI

COLLEGE OF ARTS AND SCIENCES 2010

DISCLAIMER

We are responsible for the accuracy of all opinion, technical comment, factual report, data, figures, illustrations and photographs in the article. We bear full responsibility for the checking whether material submitted is subject to copyright or ownership rights. UUM does not accept any liability for the accuracy of such comment, report and other technical and factual information and the copyright or ownership rights claims.

CHIEF RESEARCHER:

FAUDZIAH AHMAD

MEMBER:

WAN ROZAINI SHEIK OSMAN

NOR FARZANA ABD GHANI

DATE: 1 December 2010

ACKNOWLEDGEMENT

The authors wish to express their gratitude to the Ministry of Higher Education for the financial support under the Fundamental Research Grant Scheme and to Universiti Utara Malaysia for facilitating the management of the research. Gratitude also goes to the Department of Agriculture, Ministry of Agriculture and Agro-based Industry, Malaysia.

ABSTRACT

Many collective ICT initiatives and specific Internet applications has been implemented since the launch of the Multimedia Super Corridor in 1996 by the Malaysian government to offer support to the rural community. Implementing successful rural ICT applications is crucial to help bring services to the rural and also to improve the socio-economic status of the community. The agricultural sector has been given greater emphasis in Ninth Malaysian Plan, where the sector has been revitalized to become the third engine of growth in a New Agriculture.

Agribazaar is a portal developed by the Malaysian Government to provide support and assistance to the agribusiness sector through multiple agencies, led by Ministry of Agriculture and Agro-Based Industry. This initiative was implemented to help improve the agriculture industry in Malaysia. The portal has opened up more opportunities for conducting agribusiness and related industry from the local and international community. However, the review of the literature showed little work has been done on this study.

An evaluation has been carried out with the users of Agribazaar through analyzing portal user comments, interviews, and focus group discussions. This research has revealed many issues and challenges although the Agribazaar has been successfully implemented and accepted by the users. Therefore, the research provides feedback from the users to the Department of Agriculture and as well as specify the requirement for future enhancement conducted by MIMOS who is the developer.

TABLE OF CONTENT

DISC	CLAIMER	II
ACKNOWLEDGEMENT		III
ABS	ABSTRACT	
LIST	Γ OF FIGURES	VIII
LIST	Γ OF TABLES	IX
ABB	REVIATION AND ACRONYMS	X
СНА	PTER 1 : INTRODUCTION	1
1.1	INTRODUCTION	1
1.2	PROBLEM STATEMENT	3
1.3	OBJECTIVES	4
1.4	SCOPE OF RESEARCH	4
1.5	SIGNIFICANCE OF THE RESEARCH	4
1.6	REPORT ORGANIZATION	5
СНА	APTER 2 : LITREATURE REVIEW	6
2.1	AGRIBUSINESS	6
2.2	ICT IN MALAYSIA	7
2.3	INTERNET-BASED APPLICATIONS VERSUS WEB-BASED APPLICATIONS	9
2.4	PERFORMANCE EVALUATION	11
2.4.1	QUALITATIVE AND QUANTITATIVE RESEARCH	11
2.4.2	VARIABLES OF PERFORMANCE EVALUATION: ACCEPTANCE & SATISFACTION	12
2.4.3	TECHNOLOGY ACCEPTANCE MODEL	13
2.5	WEB-BASED APPLICATIONS FOR AGRIBUSINESS IN MALAYSIA	14
2.6	AGRIBAZAAR (HTTP://WWW.AGRIBAZAAR.COM.MY)	16
2.7	CHAPTER SUMMARY	19
CHA	APTER 3 : METHODOLOGY	20
3.1	PHASE I - PRELIMINARY STUDY, RESEARCH DESIGN AND DATA COLLECTION	21

3.1.1 PERFORMANCE EXPECTANCY	22
3.1.2 EFFORT EXPECTANCY	23
3.1.3 Social Influence	23
3.1.4 FACILITATING CONDITIONS	23
3.2 Phase II – Data Analyses	24
3.2.1 QUANTITATIVE METHOD	25
3.2.1. DESCRIPTIVE STATISTIC	26
3.2.1.2 RELIABILITY AND VALIDITY	26
3.2.1.3 REGRESSION ANALYSIS	26
3.2.1.4 PEARSON CORRELATION COEFFICIENT	27
3.2.2 QUALITATIVE METHOD	27
3.3 PHASE III – RESULT PRESENTATION	28

CHAPTER 4 : DATA ANALYSIS AND RESULTS

4.1	QUANTITATIVE METHOD	29
4.1.1	DEMOGRAPHIC AND DESCRIPTIVE STATISTICS	29
4.1.2	AGRIBAZAAR PORTAL USAGE	30
4.1.3	RELIABILITY AND VALIDITY OF QUESTIONNAIRE	31
4.1.4	REGRESSION ANALYSIS	32
4.1.5	PEARSON CORRELATION COEFFICIENT ANALYSIS	33
4.2	QUALITATIVE METHOD	34
4.2.1	INTERPRETIVE ANALYSIS	34
4.2.2	FOCUS GROUP DISCUSSION	37
4.3	CONCLUSION OF THE FINDINGS	38

CHAPTER 5 : CONCLUSION

39

57

29

5.1	IMPLICATION OF THE STUDY ON RESEARCH OBJECTIVES	39
5.2	PROPOSITION FOR THEORY	42
5.3	PROPOSITION FOR PRACTICE	42
5.4	LIMITATIONS AND SUGGESTIONS FOR FUTURE RESEARCH	42
REF	REFERENCES	
APPENDIX A: SCREENSHOTS OF AGRIBAZAAR PORTAL		51
APPENDIX B: ONLINE QUESTIONNAIRE		53

APPENDIX C: SPSS OUTPUT

APPENDIX D: INTERPRETIVE RESULT

LIST OF FIGURES

- Figure 2.1: Percentage of Agribazaar registered users according to business type as of 23 December 2009
- Figure 2.2: Percentage of Agribazaar registered users according to states in Malaysia as of 23 December 2009
- Figure 3.1: Research Phases
- Figure 3.2: Hypothesized Model

LIST OF TABLES

Table 2.1: The MOA's Agencies

- Table 2.2: Web-based Applications Initiative by MOA's Agencies
- Table 2.3: Agribazaar registered users according to states in Malaysia as of 13 August2007
- Table 3.1: Assumption for the relation's power between variables
- Table 3.2: Person's r indices of Correlation
- Table 4.1: Demographic Profile
- Table 4.2: Usage Information
- Table 4.3: Cronbach's Alpha reliability test for the hypothesized model
- Table 4.4: Cronbach's Alpha reliability test for constructs

Table 4.5: Summary of Hypotheses Test

- Table 4.6: Ranking of Factors according to Correlation score
- Table 4.7: Statistics on data gathered

ABBREVIATION AND ACRONYMS

Information Systems
Information Technology
Information and Communication Technologies
Technology Acceptance Model
Extension of Technology Acceptance Model
Perceived Usefulness
Motivational Model
United Theory of Acceptance and Use of
Technology
Theory of Planned Behavior
Innovation Diffusion Theory
Social Cognitive Theory
Model of PC Utilization
International Business Machines
Ease of Use
Agriculture Technology Industry-Government-
Electronic Revolution
Internet data center
Electronic Business
Electronic Government
Electronic Commerce
Theory of Reasoned Action
United Theory of Acceptance and Use of
Technology
Supply Chain Management
Malaysian Institute of Microelectronic Systems

CHAPTER 1

INTRODUCTION

1.1 Introduction

Internet based applications are currently considered as amongst the best solution to meet the requirements of the rural community due to lack of infrastructure availability and the remote locality. The Malaysian government has implemented many collective ICT initiatives and specific Internet applications to offer support to the rural, particularly the agricultural sector to improve the socio-economic status of the community. Implementing successful rural ICT applications is thus crucial to help bring services to the rural. The provision of services will help increase agricultural productivity and sustainability of the farming community.

Agriculture has long been part of Malaysia's economic sectors. However, in 1980's the focus has been diverted to industrial developments. Nevertheless, the importance of agricultural sectors to the development of the country is undeniable. This has been strongly proven when agriculture receives a very significant emphasis in the Ninth Malaysian Plan. Throughout the period, the agriculture sector will be revitalized to become the third engine of growth where emphasis will be on *New Agriculture*. This involves large scale commercial farming, wider application of modern technology, production of high quality and value-added products, biotechnology discovery, increased convergence with information and communications technology (ICT), and participation of entrepreneurial farmers and skilled workforce (Wong, 2007; Pade et al., 2006).

In line with the emphasis made on *New Agriculture*, Malaysia has moved to consider agriculture as a business. This has resulted in the formation of agribusiness. The term

"agribusiness" has existed since the mid-1950s and implies the shift from "farming as a way of life" to "farming as a business" (Sutton, 2001). The paradigm shifts have made Malaysia move one more step ahead in producing agricultural products not only for the sustainability of local food consumption but at the same time expand the economics growth development.

While agribusiness is normally associated with western multinational companies, Malaysia provides an alternative example of a Third World agribusiness (Sutton, 2001). The technology of ICT has opened up more opportunities for agribusiness to spread its wing both locally and internationally. These days agribusiness is conducted via the Web, taking advantage of the Internet technologies and facilities. Thus, it has made possible for entrepreneurial farmers to market their products outside Malaysia and increase their market share internationally.

In Malaysia, a one stop center to meet the needs of agriculture sector has been developed through the initiatives of several government agencies, where farmers are encouraged to market their produce online. Known as the Agribazaar, the online center has become one of the portals developed for the agricultural community for the purpose of buying and selling of agricultural products. Besides that, the portal was intended to enable the agricultural community to obtain, disseminate, promote, and make transactions related to agriculture and business. Since the launch in June 2004, the number of registered users had grown and had given numerous feedbacks via the portal.

The Agribazaar is an information-based portal provided by the Department of Agriculture, Ministry of Agriculture and Agri-based Industry. It is intended to offer an Internet based trade communications for buyers and sellers of farming products. Agribazaar started with 3 essentials modules which are eMake, eSupport and ePayment.

The central objective of this portal is to enhance market reach, efficiency, and productivity for farmers in the buying or selling of farming products. The application has been implemented and utilized by a considerable number of users from Malaysia and other countries across the globe.

ICT technology is believed to be a catalyst in providing more opportunities for positive expansion of agribusiness. This study begins with overview of ICT and Internet based applications, agribusiness and later discuss the role of ICT technology focusing on web based application as a platform for conducting agribusiness. Agribazaar has been chosen as a case study in this study as it is the most widely used application for the community. The portal has been designed to offer an Internet based commerce infrastructure for buyers and sellers of agricultural products (Mimos Berhad, n.d). Analysis on the implementation of Agribazaar will be presented in the later section. Thus the research aims to evaluate the portal, study its evolution and impact, and provide feedback to developers and policymakers.

1.2 Problem Statement

ICT initiative is seen as the alternative platform for the agriculture business to promote and market their services and products online. In light of this, the Government of Malaysia has implemented Internet based applications to help small scale farmers to benefit from the current Internet technology. Among the initiatives which have been undertaken by various agencies for example;

- i) AgriBazaar, which operates under AgriTIGeR (Agriculture Technology Industry Government-Electronic Revolution), intiated by MIMOS and Jabatan Pertanian.
- ii) Myfruits.org initiated by MARDI.
- iii) FAMAExchange initiated by FAMA.

The main objective of these applications is to increase farmers' productivity competitiveness as well as to bridge the digital divide. In spite of having these initiatives, many literatures have shown that farmers still face problems to market their products to customers and many are not aware of the existence of the applications (Zulkhairi et al., 2008; Elaish, 2008; Zahurin et al., 2007). This problem is compounded by the limitation of Internet access whereby the rural has access of only 14.7% as compared to 85.3% for urban (Suruhanjaya Komunikasi dan Multimedia Malaysia, 2008). Thus to ensure that the agriculture community benefits

from the ICT initiatives, the study serves to evaluate the performance of the portal in terms of utilization and identify the barriers to successful adoption of the application.

1.3 Objectives

The main objective of this study is to conduct a research on performance of Agribazaar. The study will seek to achieve the following sub-objectives;

- i) To identify existing ICT tools currently available for small scale farmers.
- ii) To examine the acceptance of Agribazaar for conducting agricultural business.
- iii) To examine the satisfactory level of Agribazaar in terms of agricultural business usage.
- iv) To examine the success of Agribazaar in terms of agricultural business.
- v) To identify issues and challenges of successful implementation of Agribazaar.

1.4 Scope of Research

Secondary data from literature and other documents and primary data has been used to evaluate the performance of Agribazaar portal based on users' perspective through a survey, portal user comments, interviews, and focus group discussions. The unit of analysis is the user who is registered with the administrator of Agribazaar. The feedback from users received through the portal has been analyzed through qualitative analysis. Developers who developed and administer the portal are interviewed and last but not least, feedback from the development teams and users in a focus group discussion session are discussed.

1.5 Significance of the Research

Although 70% of Malaysia are considered as rural, only 30% of the population stays in these areas. Studies done in Malaysia showed that the rural agricultural community does not have the same opportunity and support as the counterparts in developed countries and those who have access and connectivity (Zulkhairi et al., 2008; Elaish, 2008; Amundsveen and Solvoll, 2005). Thus by studying the implemented Internet based application performance, the findings intend to provide better understanding and feedback for developers, policymakers and related agencies to improve facilities and support for the farmers.

1.6 Report Organization

This research report consists of five chapters. Chapter 1 discusses the background and problem statement of the study. The research objectives and scope are identified also in this chapter. Chapter 2 describes related works on the study, whilst Chapter 3 presents the methodology of the research. Chapter 4 discusses on data collected and analysis. Chapter 5 presents the contribution of the research and recommendation for future work.

CHAPTER 2

LITREATURE REVIEW

This chapter presents definitions and researches related to this study.

2.1 Agribusiness

Several definitions of agribusiness are found from the literature. Stone (2005) defined agribusiness as the generation of income from the sale of a product or service or both, which facilitates the decision making of a farmer or land manager. Agribusiness has also been described as the activities that occur after harvest and prior to final sale to consumers (Stanton, 2006). Johnson et al. (2006) defined agribusiness as the sum of all operations in the economy involved in the production, processing and wholesale marketing of agricultural products. Another definition of agribusiness by the U.S Agribusiness Council refers to aspects of agricultural production, processing and distribution. This includes food, forest and fiber production including their byproduct utilization, agricultural chemicals and pharmaceuticals, agricultural finance and trade, agribusiness/farm management, agro-environmental considerations, and land development. In short, it comprises of all major elements essential to the establishment and operation of efficient agro-food enterprises (U.S. Agribusiness Council, 2006). Though all definitions stated above are true to the meaning of agribusiness, in this research agribusiness will be refereed to as any activities that are conducted to produce, distribute, market, or transport agricultural products to generate income.

According to Kinsey (1987), agribusiness enterprises in developing countries are typically in the range from small to medium-scale operations and their businesses covers from processing raw agricultural materials to providing marketing, transport, and other services.

The scenario of agribusiness in Malaysia is no different from other parts of the world. The agribusiness entities in Malaysia cover various types of agricultural products from fresh fruits and vegetables to handicraft products made of agriculture waste. The agribusiness entities have generated incomes for millions of Malaysian and have contributed to the economic growth of the country. Heading towards the sustainability of agriculture and agribusiness sectors in Malaysia, the government provides support and assistance through multiple agencies, led by the Ministry of Agriculture and Agribased Industry. The ministry has organized numerous approaches in promoting agriculture and agribusiness as a fundamental economic sector. One of the strategies is to provide an electronic platform for marketing products through Internet.

2.2 ICT in Malaysia

According to Zahurin et al. (2007), Malaysian government has identified information technologies (ICTs) as the major determinants of national development in order to achieved the proposed developed status by 2020. In addition to its impacts in moving the economy forward, ICTs are now employed as a means of reducing the socioeconomic gaps of the less privileged sections of the society.

The Malaysian government aims at establishing more community-based telecentres to achieve the coverage of all districts within the country, a project of multi-million dollars. This will break the barrier between living in the rural areas and urban areas since location will no longer be referred to as being remote. Malaysia's approach to the problem is viewed from three main perspectives: Access (making ICTs infrastructure available to all), Adoption (encouraging ICTs usage in everybody's daily life) and Inclusion (achieving social-economic value of ICTs by all Malaysians).

The digital gap can be measured by the number of people using ICTs infrastructure such as telephones, computers and internet. The recent survey jointly financed by Economic Planning Unit (EPU) of Malaysia and United Nations Development Programme (UNDP) between 2005 and 2006 measures the rate at which ICTs is reachable to households in Malaysia which was termed as the ICTs household penetration rate. The ICTs infrastructure of reference includes mobile phones, computers and internet subscription. The result of the findings showed that the digital divide/gap is more noticed in Malaysia when it comes to technologies related to information age such as computer and internet. It was revealed that the rate of computer and internet subscription per 100 households are 24.2 and 10.4 respectively on the average. For the best served areas, the penetration rate for the household computer was 59.3 per 100 households and 3.6 per 100 households for the least served areas. The internet subscription ranges from 52.8 (best mukims) to 2.2 (worst mukims) per 100 households as at year 2000.

The Malaysian economy has witnessed a series of noticeable structural changes in the last four and half decades. Immediately after independence in 1957, Malaysia has to depend on the export agricultural products mainly rubber and tin to be precise for her economy. Shortly after the independence the government invested in planting more of agricultural export-based products to survive the economy. This really assisted in the development of the country to be able to achieve the industrialization stage. Malaysian economy now depends more on manufacturing than agricultural products. As at the year of independence, the primary sectors (agriculture, forestry and mining) contributed 45% to the Gross Domestic Product (GDP) of the country while the secondary sector (manufacturing and constructions) contributed only 11% to the GDP. As a result of industrialization, there is a decline in agriculture's contribution to GDP, which as at 2000 measured 10.5% as against 37.5% contribution from manufacturing.

According to Kementerian Penerangan, Komunikasi dan Kebudayaan (2008), connecting rural people to the internet was the initiative of Ministry's Rural Internet Centres (RIC) programme and Universal Service provision (USP). The initiative being one of the earliest rural internet initiatives in Malaysia was launched on 3rd of April, 2000. This programme was designed to present a radical approach of making internet accessible to people of rural areas by looking into areas such as infrastructure needs, capacity building and content development. The private sectors (financed by government) procure the computers and internet connectivity while local post offices

provided the site, security and electricity. Training was also organized for RIC supervisors and users to enhance judicious utilization of the resources. As a way of committing people to the use of internet, a programme was formulated where a local committee will be constituted, the committee will be charged with the responsibilities of designing the community website to showcase their local products and services and in addition to provide content relevant to the local community. During the first phase, RICs were deployed to post offices in the semi-rural areas within the reach of ISDN enabled exchange. As at now, 42 RICs have been equipped with five to six personal computers connected to internet via ISDN throughout the country. According to the result of the statistics compiled in July 2004, it was revealed that more than 53,000 users now use RIC services while more than 35,000 users have undergone training under RIC programme training scheme since its inception.

2.3 Internet-Based Applications versus Web-Based Applications

The rapid growth of telecommunication and communication technologies currently has permitted various type of activities occurred virtually beyond the geographical boundary. The most common type of technology is the Internet; whereby it is extensively accepted for communication and information dissemination channel. It provides benefits to organization and even individual. By using the Internet, people can communicate at a lower cost and in faster way. Most of the organizations today are taking advantage of the opportunity with the intention of improving the organizations' overall performance. The major trend of today's Internet technology utilization is having online businesses. Business activities such as product marketing, supply chain, payment of sales transactions and customer relations are possible to be conducted over the Internet.

The World Wide Web (www or web) is always associated with the Internet technology. It is a platform for distributing software resources across the Internet, which is then presented as rich, consistent content by applications on the client (usually browser) (Evan et al., 1999). The application in this context is referring to a piece of software that sits on a personal computer and accomplishes some tasks for the user (Knuckles and Yuen, 2004).

Organizations have developed various types of web-based applications which can easily be accessed over the web-browser. The confusion between web-based applications and internet-based applications always happened everywhere. There is an often-overlooked distinct between the Web and Internet. The line between the two is often blurred, partially because the web is rooted in the fundamental protocols associated with the Internet (Shklar and Rosen, 2003).

Examples of Internet applications include e-mail, file transfer, instant messaging and www. On the other hand, a web-based application refers to an application rendered using a web-browser and run over the Internet. The web-based application communicates with web server and process data requested or provided by users. The communication between web server and client is using Hypertext Transfer Protocol (Knuckles and Yuen, 2004). Current web-based application is something more than just a "web site". A web-based application presents dynamically tailored content on request parameters, track user behaviors and security considerations (Shklar and Rosen, 2003).

The web-based application can be categorized into web portals or specific applications such as web storefronts. A web portal is a website that collects information for a group of users that have common interests (Heflin, 2003; Lausen et al., 2005). In the year 2000, free web portal services was dominated by four firms; Excite, InfoSeek, Lycos, and Yahoo. These firms dominated the majority of advertising revenues and controlled visitor traffic to various free and general interest portals (Gallaugher, 2000).

In Malaysia, various web portals have been developed by government and commercial organizations ranging from general to specialized interest. An example of a web portal for specialized interest is Agribazaar. The portal is for individuals or businesses who are interested in purchasing or selling agricultural-based products. Since this research focuses on Agribazaar as a case study, a detailed section on the portal is discussed in section 2.6.

The government of Malaysia has spent millions of Malaysian Ringgit to develop web portals for the benefit of the Malaysian society. Whether the portal has been successful or otherwise, the review of the literature revealed little work has been carried out. In fact, how effective the portal has been are not known. Evaluations on these portals are necessary in order to know the performance of the Malaysian government's initiatives program. Several approaches to evaluate performances are hence discussed in the following section.

2.4 Performance Evaluation

This section explains the concept of qualitative and quantitative research, variables used in the study and the Technology Acceptance Model (TAM) which relates to this study.

2.4.1 Qualitative and Quantitative Research

Performance evaluation proposed in this research has been carried out using mix method which combines both qualitative and quantitative analysis. This section will briefly discuss the qualitative and quantitative research relevant to this research.

Qualitative research seeks to provide understanding of human experience, perceptions, motivations, intentions, and behaviors based on description and observation and utilizing a naturalistic interpretative approach to a subject and its contextual setting (Oxford University Press, 2008). Qualitative findings grow out of three kinds of data collections; in-depth open ended interviews, direct observations and written documents. The data for qualitative analysis normally comes from fieldwork whereby the researchers spent time in the setting under study situations. Its findings may be presented alone or in combination of quantitative data. In current research practice, qualitative methods are frequently used in evaluations because they tell the program's story by capturing and communicating the participants' stories. Qualitative findings in evaluation illuminate people behind the numbers and put faces on the statistics (Patton, 2001).

On the other hand, quantitative research is based on traditional scientific methods, which generates numerical data and usually seeks to establish causal relationships between two or more variables, using statistical methods to test the strength and significance of the relationships (Oxford University Press, 2008). Quantitative research has typically been more directed at theory generation and theory verification (Punch, 2005). In the case of this research, variables used to measure the performance of internet-based application for Agribusiness in Malaysia are explained in the next section.

2.4.2 Variables of Performance Evaluation: Acceptance & Satisfaction

This section explains general concept and definition of two variables used in the quantitative analysis. They are acceptance and satisfaction of Agribazaar. The explanation of these two words is restricted to the research objectives as stated in the previous chapter.

The first word which relates to this research is acceptance of information technology (IT). It has been the subject of much research in the past two decades. Several theories have emerged that offer new insights into acceptance and use, at both the individual and organizational levels such as Technology Acceptance Model (TAM) (Lai and Li, 2005). Researchers are generally agreed that an individual's beliefs and perceptions of IT have a significant influence on their usage (Saade and Bahlee, 2005).

The concept of satisfaction has been a major subject of psychologists, and there is general agreement that satisfaction is an attitude (Pearson and Bailey, 1980; Schwab and Cummings, 1973; Wanous and Lawler; 1972). An attitude is characterized by both the direction (positive or negative) of an individual's feeling toward something and the intensity of that feeling.

In this study, the performance of Agribazaar has been evaluated based on these two variables.

2.4.3 Technology Acceptance Model

Technology Acceptance Model (TAM) is an adaptation of the Theory of Reasoned Action (TRA) to the field of Information System. TAM was developed to explain computer usage behaviors. According to TAM, perceived usefulness and perceived ease of use determine an individual's intention to use a system with intention to use serving as a mediator of actual system use. Perceived usefulness is also seen as being directly impacted by perceived ease of use.

Researchers have simplified TAM by removing the attitude construct found in TRA from the current specification (Venkatesh et al., 2003). Attempts to extend TAM have generally taken one of three approaches:

- i) By introducing factors from related models
- ii) By introducing additional or alternative belief factors
- iii) By examining antecedents and moderators of perceived usefulness and perceived ease of use (Wixom and Todd, 2005).

TRA and TAM, both of which have strong behavioral elements, assume that when someone forms an intention to act, that they will be free to act without limitation. In practice constraints such as limited ability, time, environmental or organizational limits, and unconscious habits will limit the freedom to act. Attitude towards adoption is influenced by factors such as:

- i) Perceived ease of adoption;
- ii) Apprehensiveness
- iii) Perceived utilities of technology (extrinsic motivation)
- iv) Enjoyment (intrinsic motivation)

In addition, individual characteristics like age, qualification, their prior experiences in adopting technology, technology suppliers' commitment, compatibility with existing technology and enhanced value are also important factors.

2.5 Web-based Applications for Agribusiness in Malaysia

Web-based applications for agribusiness in Malaysia are generally accomplished by government agencies. The Ministry of Agriculture and Agro-based Industry (MOA) is the key government agency which persistently providing supports via multiple channels in improving and sustaining the agriculture sector in Malaysia. The ministry performs its roles through various agencies which responsible for specific roles in supporting the agriculture sector. Table 2.1 depicts the MOA's agencies and their roles.

Agencies	Roles	
Department of Agriculture	Supporting various agricultural fields, particularly in	
	food crops and agro-processing industries to ensure	
	sufficient food production.	
Department of Fisheries	Managing and developing the fisheries sector of	
	Malaysia.	
Department of Veterinary	Managing and monitoring the animal industry and	
Services	food from animal products.	
Federal Agricultural	Coordinator of food marketing and agricultural	
Marketing Authority	resources.	
(FAMA)		
Malaysia Agricultural	Implementing research activities on food and	
Research and Development	agriculture.	
Institute(MARDI)		
Malaysia Fisheries	Improving the social-economic status of fishermen	
Development Authority	with focusing on enhancement and development of	
(LKIM)	the fisheries sectors.	
Farmer Association	Improving farmers' economic and social	
Authority(LPP)	perspectives.	
Malaysia Agricultural Bank	Providing financial assistance for the agricultural	
	activities.	

Table 2.1: The MOA's Agencies

Muda Agricultural	Supporting and managing the economic development	
Development Authority	of Muda Irrigation area.	
(MADA)		
Kemubu Agricultural	Supporting and managing the economic development	
Development Authority	of Kemubu area.	
(KADA)		
Malaysia Pineapple Industry	Managing and monitoring pineapple industry	
Board (MPIB)		

All agencies listed in Table 2.1 are vital to the agriculture growth in Malaysia. Various efforts have been carried out to ensure that the agriculture sector is constantly raised to achieve the government's visions. The current catalyst towards achieving the government's vision is by utilizing the information and communication technology (ICT). The deployment of ICT has been established as a mean to improve productivity of businesses in the other economic sectors in the world. It has been used as marketing channel for numerous types of products and services which are part of the cause to the globalization of businesses. The major implementations of the technology are the web-based applications and the mobile applications.

An early observation has been done to comprehend the web-based agribusiness applications initiative for the MOA's agencies. Table 2.2 shows the result of the observation.

Agencies	Web-based Applications
Department of Agriculture	Agribazaar
FAMA	AgrobasedProduct.com
MARDI	MyFruits.org

 Table 2.2: Web-based Applications Initiative by MOA's Agencies

Information listed in Table 2.2 shows that ICT initiatives have been done by the government to support the business activities for agriculture sector in Malaysia. One of them is known as Agribazaar, which has been chosen as a case study for this

research. The following section elaborates further on Agribazaar, a web-based application for agribusiness in Malaysia.

2.6 Agribazaar (http://www.agribazaar.com.my)

The Agribazaar is a web portal provided by the Department of Agriculture, Ministry of Agriculture and Agri-based Industry. It is designed to offer an internet-based commerce infrastructure for buyers and sellers of agriculture products (Mimos Berhad, n.d.). Agribazaar has been launched on 2003 with 3 basics modules which are e-Make, e-Support and e-Payment. The aim of this portal is to improve market reach, efficiency, and productivity among individuals or businesses involve in buying or selling agriculture products. The portal has undergone further enhancement, in terms of the number features with the inclusion of modules e-Buy/Sell, e-Logistic (Mimos Berhad, n.d.). Appendix A presents screenshots of the portal, which has remain as it is over the period of this study.

The Agribazaar is considered as a marketing channel which allows individual or organizations to register with the application in order to promote their businesses. Marketing channels can be of different types, ranging from advertising channels, order processing channels, to customer support channels (Chaudhury et al., 2001). Promoting the agricultural products is a challenging tasks due to the nature of the products itself. This requires a good strategy of marketing. Agribazaar has been designed to support the agribusiness entity via providing a platform to promote agriculture-based products and services.

The opportunity given by the Department of Agriculture (DOA) will broaden of market size of agriculture products. This is due to the fact that a web-based applications are easy to access anytime at anywhere in the world. It is proven by the number of foreign registered users of Agribazaar. These registered users are from various countries in the world which includes Bahrain, Bangladesh, China, Indonesia, India, Maldives, Mexico, Morocco, Mongolia, Nigeria, Netherlands, New Zealand, Pakistan, Switzerland, Spain, Vietnam, Sri Lanka, South Africa, Syria, Taiwan, The United States of America, the United Kingdom and some other countries in the world (Mimos Berhad, n.d.).

Based on the profile of Agribazaar users found in the portal, the number of registered users of Agribazaar is compiled. An observation has been conducted to fulfill this objective. According to Norliza et al. (2007), in the observation accomplished on 12 and 13 August 2007, the registered users of the Agribazaar increased approximately 15 users within a day. The number of registered users according to the states in Malaysia as at 13 August 2007 is presented in Table 2.3.

States Number of Users **Percentage** (%) 29.1 Selangor 7021 2772 11.49 Kuala Lumpur Johor 2303 9.55 Terengganu 1778 7.37 Perak 1725 7.15 Kedah 1445 5.99 Kelantan 1203 4.99 1205 4.99 Pahang **Pulau Pinang** 999 4.14 Negeri Sembilan 903 3.74 695 Melaka 2.88 599 2.48 Sarawak 459 1.9 Sabah 1.74 Others 420 Putrajaya 261 1.08 Perlis 243 1.01 94 Wilayah Persekutuan Labuan 0.39

Table 2.3: Agribazaar registered users according to states in Malaysia as at 13August 2007

Total Number of Users	24125	100

According to Table 2.3, the state with the highest number of users is Selangor with 7,021 agribusiness entities which contributed to 29.10% of the total users. It is then followed by Kuala Lumpur and Johor with 11.49% and 9.55% respectively. The state with the lowest number of users is Wilayah Persekutuan Labuan with 0.39%. In total there were 24,125 registered users in the entire country and 420 from the other countries in the world.

The number of registered users has increased gradually over the period of this study. As reported in the portal (Mimos Berhad, n.d.), the total of registered number of users as at 23 December 2009 is 55,632 members. Back in 2007, the total of registered user reached at 28,371 members and increased to 41,737 members in 2008. Collectively, within three years, the average number of newly registered user per month was between 1,100 to 1,160 users. They can be classified according to business types or subsectors, as presented in the Figure 2.1. The proportion of users according to states in Malaysia as at 23 December 2009 stood as similar as back in 2007, as depicted in Figure 2.2 with Selangor, Kuala Lumpur, Terengganu, and Johor to be the top four states, where most of Agribazaar's users are located.



Source: Agribazaar Portal

Figure 2.1: Percentage of Agribazaar registered users according to business type as at 23 December 2009



Source: Agribazaar Portal



2.7 Chapter Summary

Agriculture business is one of key economic sectors in Malaysia which has contributed to a great amount of revenue to Malaysians. The advancement of information and communication technology (ICT) drives the traditional method of agriculture business to the borderless transactions. As a result, Agribazaar has been setup by Government of Malaysia to be a platform for agriculture traders to conduct business electronically. As it has been the dominant portal for conducting business among agriculture traders for more than seven years, it is important to measure the performance of such internet-based application and its effectiveness to users, society and national economy.

CHAPTER 3

METHODOLOGY

This chapter describes the research design, research hypothesis and methodology used in this study. It includes descriptions on the sampling of the study, survey instrument, data collection procedures and statistical techniques for data analysis. This research is an exploratory research study as its major purpose is to gain a better understanding of the performance and acceptance of Agribazaar portal amongst agriculture community. The research has been conducted in several phases. Figure 3.1 presents the approach used.



Figure 3.1 Research Phases

3.1 Phase I - Preliminary Study, Research Design and Data Collection

The research begins with preliminary study from which the factors or constructs for determining the use and acceptance of Agribazaar are drawn from. The research framework for this study is adopted from Venkatesh et al. (2003). This framework has been popularly referred by other researches and has been used to evaluate internet websites (Fritz & Schiefer, 2003; Fritz & Schiefer, 2002).

Technology Acceptance Model (TAM) has also been incorporated in the framework to perform the analysis. The motivation behind the integration of technology acceptance research with the criteria identified is to have a better understanding of the acceptance and use of internet website specifically internet based applications for agribusiness.

There are many factors as suggested by Venkatesh et al. (2003), Lesley University Library (2007) and Schrock (2006) to determine the use and acceptance of internet website. Most of which can be grouped into four major constructs: Performance Expectance, Effort Expectancy, Social Influence and Facilitating Conditions. Since these constructs have been largely used for related research works, this research uses the same four constructs. These constructs are hypothesized to directly influence User Acceptance. When employed together these constructs are identified as the factors that determine the technology use.

Behavioral Intention has been omitted as suggested by Ajzen (1991) and Straub, Limayem & Karahanna (1995). The reason lies in the role of behavior itself, or the use of the system is more important than the intention to use it. Therefore, the most important aspect of this study is to find out whether the users actually use the technology and not concentrating on their intention towards using it.

Figure 3.2 represents the research framework used in this study.



Figure 3.2: Hypothesized Model

The following sections explains the constructs used.

3.1.1 Performance Expectancy

Based on many website evaluation criteria from various researches (e.g. Fritz & Schiefer, 2003; Fritz & Schiefer, 2002; Lesley University Library, 2007; Schrock, 2006), performance expectancy plays important role in influencing the use of a website. Some of the related criteria which falls under performance expectancy such as the comprehensiveness and the reliability of the information, whether or not the use of the agribusiness website can increase the productivity (Lesley University Library, 2007; Schrock, 2006), and whether or not the use of the agribusiness website can increase the productivity (Lesley University Library, 2007; Schrock, 2006), and whether or not the use of the agribusiness website can increase the chance of getting business (Fritz & Schiefer, 2003). Performance expectancy therefore can be described as the confidence level of a person to increase his or her job performance when using a system (Venkatesh et al., 2003). When most of the criteria meet the expectation of users in using agribusiness website, users will feel confident that they will get business and meet trusted people on the website to do business with.

3.1.2 Effort Expectancy

This construct is related to how far users believe that the website is easy to use (Venkatesh et al., 2003). Some criteria including perceive ease of use (Davis, Bagozzi & Warshaw, 1989), complexity including coverage and clarity of the website content, and ease of use including the accessibility and availability of the information on the website (Lesley University Library, 2007). The ease of use of agribusiness website is important in order not to give difficulty and hamper users, especially the inexperience users to use it. Criteria evaluated on Agribazaar portal includes whether or not working with the system is complicated, the clarity of the information on the website, whether or not the website load easily and quickly, and whether or not it is free to obtain the information from the website.

3.1.3 Social Influence

According to (Venkatesh et al., 2003), social influence is "defined as the degree to which an individual perceives that important others believe he or she should use the system". This construct describes the individual's behaviour, influenced by how the surrounding people will judge him or her when they are using a particular technology. Some criteria employed in evaluating Agribazaar portal includes whether or not there are people who are important to the user encourage the use of the website, whether or not any organization support the use of the website, whether or not it will improve the credibility/reputation and position of the user, and whether or not it will improve the interactivity between seller/buyer over the websites at anytime.

3.1.4 Facilitating Conditions

This construct relates to how far an individual believes the existence of proper technical and organizational structure to support and maintain the system (Venkatesh et al., 2003). In agribusiness website context, this factor is important to determine the reputable organization is there behind the system to support it (Fritz & Schiefer, 2003; Fritz & Schiefer, 2002). Another aspect is the attractiveness, design and navigation system of the website, whether or not the menu items are logically grouped and the

website is compatible with all matters related to individual's work (Schrock, 2006). In the context of agribusiness website, for example, whether or not the website offers facility to categorize all agriculture products advertised in the selling page, according to easy-to-understand categories.

Based on the constructs several hypotheses have been made. The hypotheses are required to test the assumptions of the influential factors. The following are the hypotheses:

- H1: Performance Expectancy will positively influence Agribazaar portal use.
- H2: Effort Expectancy will positively influence Agribazaar portal use.
- H3: Social Influence will positively influence Agribazaar portal use.
- H4: Facilitating Conditions will positively influence Agribazaar portal use.

This study has been conducted amongst the Agribazaar portal users and data has been collected by means of a survey conducted online in Malaysia in 2008. Convenient sampling has been used for data collection. A total of 1200 questionnaire forms (Appendix B) have been delivered to respondents of which 119 were returned giving a response rate of 10%. To exclude incomplete and inappropriate questioners a simple cleansing method was used and this resulted in a sample that was well distributed in terms of demographic information (e.g. age, and education level) was used for data presentation and analysis.

3.2 Phase II – Data Analyses

Two methods of analysis have been used in the study. Quantitative method is used to perform various statistical measures. The measures aim to identify the influential factors of Agribazaar usage. The qualitative method has been applied on analyzing the feedbacks obtained from users in the portal and focus group discussion. The qualitative findings are to support the numerical analysis. The following sections explain both methods.

3.2.1 Quantitative Method

Quantitative method such as descriptive statistics, regression, and correlation have been used on data collected through questionnaires. The questionnaires were distributed to a broad range of users in the portal to create variances in terms of gender, age, educational level and the portal's usage experience.

The survey instrument has been developed based on constructs identified from literature review (Venkatesh et al., 2003; Lesley University Library, 2007 and Schrock, 2006). A pilot test with the first group (5 participants) was conducted to ensure the content validity of the measures is applicable in the current context. Based on the feedback, the questionnaire has been revised accordingly. The updated questionnaire has then been validated with a second group of pilot test participants (5 participants) and feedback has been requested. No significant suggestion has been received and therefore, no further changes have been made. Subsequently, the questionnaire has been distributed to users by sending an invitation email prompting them to respond in a dedicated website.

All determinants are measured using a scale adopted from study by Dennis, Venkatesh & Ramesh (2007). In addition to the above measures, gender, age, educational level and Agribazaar portal usage experience are measured using single items, which are acceptable given the factual nature of these constructs. The questionnaire (Appendix B) consists of two parts: Part 1 requests the respondents to provide some background information e.g. age, gender, educational level, monthly income and years of experience using Agribazaar; and Part 2 consists of 21 questions on a 6-point Likert scale (ranging from 'strongly agree' (coded 5) to 'strongly disagree (coded 1) and 'not applicable' (coded 6)), which represents the items to measure the awareness and satisfactory level of using Agribazaar.

Respondents were given ample time to answer the questionnaire. In each phase of the survey, the timeframe is set to one month to complete the questionnaire. The survey was administered on a dedicated website and the data were kept in a database. This is to ensure that, the respondents can access the questionnaire conveniently and are

recruited from a random list of contacts obtained from Agribazaar portal. After collecting all the responses, a statistical software package is used to analyze the data.

The following section describes the statistical measures used for data analysis.

3.2.1.1 Descriptive Statistic

The important role of statistics is data gathering, formulating and providing formulation that can be easily understand by projecting group. Descriptive statistic can be used to describe the sample pattern. As mentioned earlier, this study is set to test the hypotheses, in order to determine which factor is most related. Every factor's Mean is calculated to measure the importance of each of them respectively.

3.2.1.2 Reliability and Validity

The first step is to examine the reliability and validity of the questionnaire. Reliability and validity tests are conducted to determine the reliability of the questionnaire. According to Sekaran (2000), the value of Cronbach Alpha range between the coefficient value of .70 to .98 shows that the questionnaire is considered highly reliable. On the other hand, if the coefficient value of reliability is lower than .60, the questionnaire has low reliability and must be rejected (Sekaran, 2000). Therefore, given the exploratory nature of this study, .70 is chosen as the cut-off point for the reliability test.

3.2.1.3 Regression Analysis

The regression analysis was chosen to test the proposed hypotheses in this study and analyzed how dependent variables can be used to predict a dependent linear regression variable based on the correlation between the variables. The nullhypothesis is rejected, if the p-value is lower than .05. The regression scores for each of the dependent variables are categorized as high, medium and low. Assumption for the relation's power between variables can be seen in Table 3.1.
Regression Coefficient	Relation's Power
.0020	Can be disregarded
.2040	Low
.4060	Middle
.6080	High
.80 -1.00	Very High

 Table 3.1: Assumption for the relation's power between variables

3.2.1.4 Pearson Correlation Coefficient

This correlation measure is used to further examine how much the dependent variable selected correlate with Actual Use. The scale used in this study has been suggested by Sekaran (2000), which is used to describe the intensity of relationships between the dependent and the independent variables of the study as shown in Table 3.2.

Pearson's r	Indication
Between ± 0.80 to ± 1.00	High correlation
Between $\pm~0.60$ to $\pm~0.79$	Moderately high correlation
Between $\pm~0.40$ to $\pm~0.59$	Moderate correlation
Between ± 0.20 to ± 0.39	Low correlation
Between ± 0.01 to ± 0.19	Negligible correlation

Table 3.2: Person's r indices of Correlation

3.2.2 Qualitative Method

In addition to the quantitative method, interpretive analysis has been employed by analyzing the comments posted by the users in the User Feedback section on the Agribazaar portal. In order to conduct the interpretive analysis, the comments and feedback has been obtained freely and no permission from the portal administrator is needed to obtain the data. A specific date is set, as the comments kept increasing each day. As at 8 August 2007, 372 comments were extracted and analyzed. The comments

have been taken after deducting the duplicate entrees, and irrelevant comments. Issues raised by the users have been identified.

The interpretive analysis has been employed to further understand the Agribazaar portal's user and their point of view when using the portal. According to Kaplan & Maxwell (1994) as cited in Myers (1997), the goal of interpretive analysis is to understand the phenomenon from the point of view of the participants and its particular social and institutional context, when the quantified textual data is unable to explain this. In addition, qualitative findings in evaluation put people in perspective and take consider the people's point of view to support the numbers (Patton, 2001).

Therefore, in order to further support overall findings, the study also employed feedbacks obtained from a focus group discussion to verify the quantitative and interpretive findings. The discussion was organized by MIMOS on 18 and 19 November 2009 in Port Dickson, Negeri Sembilan. The workshop was attended by the officers from the related agencies of Ministry of Agriculture, selected users of Agribazaar according to business types defined in the portal (e.g. IKS/SMI and Manufacturer, Retailer and Supplier, Farmer and Breeder) and was facilitated by MIMOS. The main objective of the workshop was to conduct a post-mortem of the existing Agribazaar portal and as well as to get feedbacks from the users for future enhancement of the portal.

3.3 Phase III – Result Presentation

In this phase, the result from Data Analyses are interpreted and presented. The results aim to answer the all research objectives as mentioned in Chapter 1. Details of the result are presented and discussed in Chapter 4.

CHAPTER 4

DATA ANALYSIS AND RESULTS

In this chapter, activities conducted for Phase I and results of analysis conducted in Phase II of the research methodology are presented.

4.1 Quantitative Method

4.1.1 Demographic and Descriptive Statistics

The statistics of the total registered users over the period of this study has been discussed in detail in Chapter 2. The current number of registered users of Agribazaar has increased gradually to a hefty number of 55,821 members as of 4 January 2010. A survey was conducted online from 1st of June until 31st of July amongst Agribazaar users in 2008. The first phase was conducted from 1st June until 30th June, 2008. However the response rate was very low with only 1.8%. Only 35 responses recorded from 1876 invitations sent via email, prompting the users to answer the questionnaire. To overcome this issue, the second phase was carried out from 1st of July until 31st of July, 2008 with different approaches such as sending out new invitation email in Bahasa Malaysia version in order to increase response rate and sending out the invitation to primary email addresses, excluding secondary email addresses such as the one registered under Agribazaar portal.

As a result, the response rate increased to 10%. The number of usable respondents has increased to 119, with 84% of the respondents being male and 16% of the respondents being female. Majority of the respondents are between 31 and 40 years of age (46%). In terms of their educational level, majority of them have completed their Bachelor degree (34%). In addition, from the respondents who are working (N=104), 40% of

them work in the private sector and earn around RM1,001 – RM3,000. The demographic profile of the respondents is shown in Table 4.1.

Demographic profile	Frequency	Percentage (%)
<u>Gender</u>		
Female	19	16
Male	100	84
<u>Age (in year)</u>		
Below 21	0	0
21-30	36	30
31-40	55	46
41-50	19	16
Above 50	9	8
Education level		
PMR	1	0.8
SPM	24	20
STPM	9	8
Bachelor Degree	40	34
Master Degree	12	10
Others	33	28
Organization's Type	• •	
Government	20	19
Semi-Goverment	10	10
Private	41	40
Personal Business	12	11
Not Mentioned	21	20
.		
Income Level	0	0
KM 1,000 and below	0	0
KM1,001 – KM3,000	42	40
RM3,001 – RM5,000	30	29
Above RM5,001	11	11
Not Mentioned	21	20

Table 4.1: Demographic Profile

4.1.2 Agribazaar Portal Usage

The respondents have been enquired on the usage of the portal, in terms of years of experience using Agribazaar. Majority of respondents have between 1 to 3 years of experience (39%). The internet (email, search engine and links from other website)

plays an important role as the main marketing channel of Agribazaar as 42% users have revealed that they first knew about the portal from this channel. Meanwhile, Brochure and Magazines play the least role in this context with only 2% of users. A number of 104 respondents (87%) are registered users while the remaining is not. Table 4.2 shows the statistics.

Table 4.2: Usage Information

Agribazaar usage	Frequency	Percentage (%)
Experience using Agribazaar (in year)		
Less than 1	25	21
1-3	46	39
4-5	4	3
More than 5	3	3
Cannot be determined	41	34
Registered user		
Yes	104	87
No	15	13

4.1.3 Reliability and Validity of Questionnaire

For the data analysis, Cronbach Alpha reliability coefficients for the questionnaire (20 indicators) and each construct has been used. Results are shown in Table 4.3. According to Sekaran (2000), Cronbach Alpha's value of 0.70 and above is acceptable, and lower than 0.70 indicates an unreliable scale. A value of 0.866 has been obtained for the test which shows that all 20 indicators used in this study seems reliable.

Table 4.3:	Cronbach's Al	bha reliability	v test for the	hypothesized	model

Cronbach's Alpha	Cronbach's Alpha Based On Standardized Items	No of Items
.866	.872	20

Cronbach's Alpha reliability test has also been conducted on the constructs. Table 4.4 presents the results. From the table, it can be seen that the four constructs obtained values of more than 0.70 indicating that the reliability of all measurement scales was above the recommended minimum level of 0.70.

Constructs	Cronbach's Alpha	No. of Items
Performance Expectancy	0.782	5
Effort Expectancy	0.823	5
Social Influence	0.793	5
Facilitating Condition	0.742	5

Table 4.4: Cronbach's Alpha reliability test for constructs

4.1.4 Regression Analysis

Due to the limited sample size, statistical analysis was limited to correlation and regression analysis. Correlation analysis results appear in Table 4.6 which includes the observed correlations. Table 4.5 also depicts the results and the associated p-values in accordance with the research model.

The regression analysis has been used to expose how different factors affect the use of Agribazaar portal. Statistically, the model used for this study is statistically significant when the p-value is .000, which is less than .001. The results of the regression analysis are presented in Appendix C. Table 4.5 depicts the summary of the result obtained.

Table 4.5: Summary of Hypotheses Test

Hypotheses Tested	Coefficient	Supported? Y/N
H1	0.214	Y
H2	0.282	Y
H3	0.242	Y
H4	0.267	Y

User acceptance to utilization of Agribazaar is positively affected by Performance Expectancy (PRE) (β =0.214 P<0.01) thereby supporting hypothesis one. This indicates that utilization of the portal will increase if an individual believes that using the system will help him or her to get business leads and increase productivity, thus help them to increase sales.

In terms of Effort Expectancy (EFE), result in the Appendix C shows that user acceptance to utilization of the portal is positively affected by effort expectancy (β =0.282 P<0.01) thereby supporting hypothesis two. This indicates that utilization of Agribazaar will increase if the users feel the portal is easy to use thus making it easier to market their products. Another aspect is when the navigation of the system is simple, the users will feel more likely to use the portal.

The users also feel that their credibility will improve by using the website. User acceptance to utilization of Agribazaar is positively affected by Social Influence (SOE) (β =0.242 P<0.01) thereby supporting hypothesis three. This indicates that utilization of Agribazaar will increase if the users perceive that others believe he or she should use the portal to market their products. On another aspect, the interaction between seller and buyer on the portal also influence the utilization.

Facilitating Condition (FCC) also positively affects the utilization (β =0.367 P<0.01) thereby supporting hypothesis four. The utilization of Agribazaar will increase if the user believes that an organizational and technical infrastructure exists to support use of the portal. With the support from the DOA and MIMOS, the users feel more confident to use the portal. In summary, the result from this analysis shows all hypotheses are supported. Effort Expectancy had the strongest effect with a path coefficient of 0.282 emphasizing the important role of the user friendliness of the portal in driving his/her acceptance toward using Agribazaar.

4.1.5 Pearson Correlation Coefficient Analysis

This study runs a correlation analysis to further test the hypotheses. The results indicate that all the purposed factors are positively correlated with Actual Use. The score of importance of factors are shown in Table 4.6 and Appendix C. Other factors

such as background variables (gender, age, educational level and income level) showed low correlation with use and therefore have been disregarded in the model.

Factor	Correlation score
EFE	.831
SOE	.818
PRE	.786
SOC	.730

Table 4.6: Ranking of Factors according to the Correlation score

4.2 Qualitative Method

This section presents results obtained from interpretive analysis and focus group discussion.

4.2.1 Interpretive Analysis

Based on the information gathered from users feedback found in the portal, interpretive analysis has been conducted. Table 4.7 shows the statistics of data gathered as of 8 August 2007.

Description	Total
Comments	590
Duplicate Entries	20
Irrelevant Comments	217
Entries Analyzed	353
Comments Analyzed	372
Number of Issues Raised	32

Table 4.7 :	Statistics	on data	gathered
--------------------	-------------------	---------	----------

From the analysis of the total comments (N=372) a number of issues can be highlighted:

i) Account accessibility.

23.66% of users could not access their account. Most of them could not access their accounts due to incorrect login id or password. There is no facility in the portal to retrieve forgotten logins or passwords. A few users could not login even when the correct login ids and password have been used.

ii) Portal Content.

10.75% of the users found that the portal provides inadequate information. Users suggest various agricultural/farming materials to be included in the portal. These are:

- Herbs for traditional medication
- Landscaping plants
- Product packaging and marketing services
- Pesticide
- Aquatic products
- Daily market price for vegetables and fruits
- Agricultural techniques and technologies
- Horse breeding
- Hydroponics planting
- Mushroom planting
- More categories in the tender/offer section
- Recipes trading in general
- Training courses
- Advertisements on agricultural products or services

Further feedback from the users also raised the issue of knowledge sharing amongst the registered users as currently there is no Forum or Chat feature to communicate and share ideas between the interested parties. Hence, the users feel less interactivity on the portal.

Furthermore, the legitimate company registration information for a registered user in most cases is unavailable under each user profile. According to the findings, company registration number is important to determine between individuals whom are interested to do business, or individuals whom are casually registered with the portal and not actively or directly involved in getting business leads from the portal.

iii) Registration.

9.41% of the total comments are related to registration problem. Among the issues concern are registration could not be done using the 3G phone, international candidate could not register as he is without an identification number, users receive no confirmation/ acknowledgement after registration, application rejected, unsuccessful registration, compulsory columns could not be filled, and waiting period after registration is too long as user can only receive the login id and password after 3 days of registration.

iv) Software features.

4.84% of the users highlighted their concern on the efficiency of the software. Most of them felt that the loading of pages is slow reaching up to 10 minutes for one page at certain times. Several suggestions have been made to upgrade the services. These include offering after office services, increase categories on products/services displayed on the website, user friendlier interface, speed up response from web master or web administrator, and encourage users to be more active in discussion or interaction.

v) Emails accessibility.

4.30% of the users have problems with their emails. Users could not access their emails, emails could not be sent to the portal administrator and users receive many spam emails.

vi) Information accuracy.

General information such as contact details e.g. telephone and fax numbers and address of the State Agriculture Department is not up to date. The respondents (3.49%) also highlighted the needs to filter old postings in the portal especially in comment and e-Buy/Sell sections.

vii) Portal promotion.

3.49% users highlight the issue of portal promotion. Users suggest the government to promote the website extensively as in their opinion; there are many people who are unaware of the portal existence. Users perceived that there will be more activities/offerings for selling and purchasing when the portal is publicly and widely known.

viii) Other issues

Other problems highlighted by the users consist include non user-friendliness interface, language barrier as all offerings are mostly in native language (Bahasa Malaysia), difficulty to edit the postings e.g. selling and purchasing, unorganized website content, static information on the website, difficulty in searching for specific information, lack of pictures, computer illiteracy amongst the users, and broken links.

Nevertheless, 15.32% of the users agree that the Agribazaar portal provides useful information such as product directory for trading activities. Examples are information on contacts of prospect buyers or sellers. Appendix D represents result of the interpretive analysis based from the comments taken for from the portal.

4.2.2 Focus Group Discussion

In addition to the findings obtained through interpretive analysis, the study also discovered interesting facts related to the usability problems of the Agribazaar portal, obtained from a focus group discussion. It has been found that findings obtained from the discussion supported the findings from the quantitative and interpretive analyses, in terms of usability aspect of the portal.

Technical issues such as slow page loadings, complicated registration process, are too cluttered pages with information and unrelated images, and password recovery problem have also been highlighted by the users. In addition, it has been found that only the e-Buy/Sell feature is preferred. Other features such as e-Stock, e-Logistic, e-Support and e-Make have not been utilized.

Privacy issue has been another important issue highlighted by a user. Examples of privacy issue include dissemination of personal details, such as full name, telephone number, and address of the business owner. The portal has no control of exposing users profile to the public. Collectively, all members in the focus group agree to the fact that the DOA has never promoted the portal extensively to the public, which results to little awareness on the existence of the portal.

The role of portal administrator was also queried in terms of slow and sometimes no response to requests and complaints made by users. As a result, users are not able to get a new password if they forgot their existing password, and thus, could not personalize their offers in the e-Buy/Sell section. For example, to add a new category in the e-Buy/Sell offer form, a user need to wait for the portal administrator to approve the new category first before they can post a new offer on the portal. Without a password they are not able make a request to the portal administrator and thus, could not post a new offer.

The focus group members also agreed that some important information such as company's registration number is unavailable causing problems in making business deals. Chat and forums are also two important features that are missing in the portal, thus making Agribazaar to be less attractive to potential business users.

4.3 Conclusion of the findings

Conclusion and implications of the overall findings will be discussed in Chapter 5. The final part of the conclusion will also touch on the limitations and suggestions for future research.

CHAPTER 5

CONCLUSION

This chapter discusses findings of the study, highlights issues and provides recommendations.

5.1 Implication of the Study on Research Objectives

The main objective of this study is to evaluate the performance of Agribazaar. The study seeks to achieve the following sub-objectives;

i) To identify existing ICT tools currently available for small scale farmers.

Due to the limitation of funds, data to fulfill the first objective has been collected from secondary sources. Particularly, from researches carried out by Zulkhairi et al. (2008); Elaish (2008); and Zahurin et al. (2007) in the rural northern region of Malaysia. The studies found that the agriculture community do not have high education background, and have little ICT knowledge. The highest education level is SPM. This is expected as most of the young people who are educated had left their villages to study in the higher institution of learning in towns and cities. The people who are involved in the agriculture sector comprises mainly of the remnants of the uneducated youth and older members of the community.

The findings obtained from this showed that the percentage of farmers and breeders using Agribazaar are 22.3% and 10.8% respectively forming a small portion of the users using the portal (<u>www.agribazaar.com.my</u>). The majority of the respondents (42%) revealed that they knew about the portal from surfing the internet, whilst a mere 2% discovered the portal from brochure and magazines.

This findings has been supported by Zulkhairi et al. (2008) and Elaish (2008) which stated that less than 10% of farmers in the paddy, rubber, fishing, and tropical fruits sectors in selected villages have heard about Agribazaar. In fact, less than 50% of the Agriculture College students interviewed in FELDA Bukit Tangga have heard of the portal. Thus it can be concluded that the awareness level amongst the agricultural community is still very low.

ii) To examine the acceptance of Agribazaar for conducting agricultural business.

The performance expectancy, effort expectancy, social influence, perceived facilitating condition has positive effect on user acceptance for utilization of Agribazaar's portal. This shows that respondents have accepted the web based application for marketing their products. The four constructs of utilization of Agribazaar accounted for more than 59.2% of the variance in this variable. This indicates that the four constructs are acceptable measures to evaluate actual use of Agribazaar.

iii) To examine the satisfactory level of Agribazaar in terms of agriculture business usage.

The portal have not been upgraded or enhanced since it was implemented six years ago. Thus there have been many dissatisfactions and limitations of the application. The current version of the portal is slow, and unfriendly. Mainly, usability issues have been the concerns of users. The comments posted on the portal showed that users have been unhappy with the usability of the portal and services facilitated by the administrator. These complaints have also been raised in the focus group discussion.

The new version of Agribazaar is in the pipeline as the Department of Agriculture who is the owner of the portal and together with the developer, MIMOS have decided to upgrade the application in order to address the dissatisfaction and limitation provided by the users. iv) To examine the success of Agribazaar in terms of agriculture business usage.

Feedback from the focus group discussion revealed that the portal has created business leads. Through the portal, customers contacted business entrepreneurs via email or telephone after posting information on their products or offerings. Although Agribazaar was designed to allow buying, and selling online, the portal only allows information of the products to be posted, while the transactions occurs through secondary channels such as email, telephones or face-to-face.

Thus it can be seen that the portal is still in stage 1 of an e-Business model. According to Applegate et al. (2002) in stage 1, sellers are able to go direct to the customers without having to go through the middleman. However, in spite of the limitations, users of the Agribazaar portal have grown from 28,371 in 2007 to 55,821 members as of January 2010. This shows that the web based application has allowed farmers and other related agencies to increase their business opportunities.

- v) To identify issues and challenges of successful implementation of Agribazaar.
 - (a) Most of the users have been unhappy with poor systems support. 23.66% of the users commented on their account inaccessibility due to account being blocked or slow feedback obtained from the system administrator.
 - (b) Technical issues on the system have also been highlighted. These include slow page loadings, inaccessibility to emails, and cluttered pages.
 - (c) The accuracy of the information in the portal is a concern for most users. One example is the contact details, telephone or address of officers in charged is not updated. In addition, old postings are not filtered in the comment and e-Buy/Sell section.
 - (d) The e-Buy/Sell feature has been preferred but not e-Stock, e-Logistics, e-Support, e-Plan, e-Payment, and e-Make.
 - (e) Forum or chat section to discuss and share knowledge amongst users is not available.

(f) Little awareness on the portal existence. The Department of Agricultural has done little effort in promoting the portal and provides inadequate training sessions to targeted users.

5.2 **Proposition for Theory**

This study validates four constructs as proposed by Venkatesh (2003). The model explains 59.26 % of the variance in user acceptance of Agribazaar's portal. Thus, the findings will expectantly spark more research into factors that influence adoption of other technological innovations.

5.3 **Proposition for Practice**

This study has shed light on some of the major factors which influence user acceptance for utilization of Agribazaar portal. The findings serve as feedback to the developer of Agribazaar and the Department of Agriculture. There is a need to promote the portal and provide training to targeted users such as farmers and middlemen.

A number of factors have been identified to be the reason for the low number of users. Among the factors are the diversity of agriculture sub-sectors, the number of people involved in agriculture, ICT literacy rate among the people involved in agriculture sectors, ICT infrastructure and promotional activity of Agribazaar.

5.4 Limitations and suggestions for future research

The sample used in this study consisted of a survey conducted online in 2007. In future, it is recommended that a research on determining factors affecting the actual usage of Agribazaar portal among users to be conducted. Additionally, the applicability of UTAT constructs in the context of needs can be explored.

Respondents in this study evaluated their usage intentions based on their perceptions of the portal and as a result factors that have been found are less significant. Other factors such as Facilitating Condition may be significant if subjects were surveyed for actual usage. A study using a pre-implemented and post-implemented method would be another way of analyzing the successfulness of the portal.

Finally, other factors affecting user acceptance towards the utilization of Agribazaar may exist and further investigation on new factors is most desirable. Further testing and expansion of the model in this study thus, may capture factors not contemplated herein.

REFERENCES

- Ajzen, I. (1991). "The Theory of Planned Behaviour," *Organizational Behavior and Human Decision Processes*, vol. 50, pp. 179-211.
- Ajzen, I. and Fishbein, M. (1980), Understanding Attitudes and Predicting Social Behavior, Prentice-Hall, Englewood Cliffs,NJ.
- Applegate, L.M., Austin, R.D. and McFarlan F.W. (2002). Creating Business Advantage in the Information Age, 1st Edition, Boston: McGraw-Hill/Irwin.
- Chaudhury, A., Mallick, D.N. and Rao, H.R. (2001). Web channels in e-commerce. Communications of the ACM, 44(1), pp.99-104.
- Davis, F. D., Bagozzi, R. P. and Warshaw, P. R. (1989). "User Acceptance of Computer Technology: A Comparison of Two Theoretical Models," *Management Science*, vol. 35, pp. 982-1002.
- Davis, F.D. (1989), "Perceived usefulness, perceived easy of use, and user acceptance of information technology", *MIS Quarterly*, September, pp. 319-40.
- Davis, F.D. (1993), "User acceptance of information technology: system characteristics, user perceptions and behavioral impacts", *International Journal of Man-Machine Studies*, Vol. 38, pp. 475-87.
- Davis, F.D., Bagozzi, R.P. and Warshaw, P.R. (1989), "User acceptance of computer technology: a comparison of two theoretical models", *Management Science*, Vol. 35 No. 8, pp. 982-1003.
- Davis, F.D. and Venkatesh, V. (1996), "A critical assessment of potential measurement biases in the technology acceptance model: three experiments", *International Journal of Human-Computer Studies*, Vol. 45, pp. 19-45.

Dennis, A. R., Venkatesh, V. and Ramesh, V. (2007). "Adoption of Collaboration

Technologies: Integrating Technology Acceptance and Collaboration Technologies Research," 2003, pp. Retrieved 3 Mac 2007, from www.indiana.edu/~isdept/research

- Elaish, M. M. A. (2008). The Usefulness and Ease of Use of WAP Appication for Local Agriculture Product (Unpublished master's thesis). Universiti Utara Malaysia, Sintok.
- Evan, M.P., Phippen, A.D., Mueller, G., Furnell, S.M., Sanders, P.W. and Reynolds,
 P.L. (1999). Strategies for Content Migration on the World Wide Web. Internet Research: Electronic Networking Applications and Policy, 9(1), pp. 25-34.
- Fishbein, M. and Ajzen, I. (1975), Belief, Attitude, Intention, and Behavior: An Introduction to Theory and Research, Addison-Wesley, Reading, MA.
- Fritz, M., Schiefer, G. (2002) Market Monitoring in Dynamic Supply Networks and Chains: an Internet-Based Support System for the Agri-Food Sector. Journal on Chain and Network Science 2 (2): 93-100
- Fritz, M., Schiefer, G. (2003). Identification and evaluation of Internet resources for agribusiness information needs. In: Proceedings of the 4th Conference of the European Federation for Information Technology in Agriculture, Food and Environment (EFITA 2003). Debrecen - Budapest, Hungary, 5-9 July, 2003. Vol.1, pp. 346-351
- Gallaugher, J.M. (2000). Portal Combat: An Empirical Study of Competition in the Web Portal Industry. *Journal of Information Technology Management*, 11(1-2). pp.13-24.
- Heflin, J. (Ed.). (2003).Web Ontology Language (OWL): use cases and requirements'', W3C working draft, 31 March, Retrieve August 2, 2007 from www.w3c.org

Johnson, M., Malcolm, B., and O'Connor, I. (2006). "The Role of Agribusiness Assets in Investment Portfolios". Australasian Agribusiness Review - Vol. 14. Retrieved Nov 7, 2006 from <u>http://www.agrifood.info/review/2006/Johnson_Malcolm_O%27Connor.html</u>

Kaiser, H. F. (1974), 'An index of factorial simplicity', Psychometrika, 39 (1), 31-36.

Kementerian Penerangan, Komunikasi dan Kebudayaan. (2008). Retrieve 23 November, 2008 from <u>http://www.aspirasidigital.net.my/MainTemplate02.asp?contentid=3</u>

Kinsey, B. H. (1987). Agribusiness and Rural Enterprise, New York: Croom Helm.

- Lai, V.S. and Li, H. (2005). Technology acceptance model for internet banking: an invariance analysis. *Information & Management*, 42(2), pp. 373-386.
- Lesley University Library. (2007). "Evaluating Web Sites,", pp. Retrieved 23 September 2008, from www.lesley.edu/library/guides/research/evaluating_web.html
- Liechti, O. (2000). Awareness and the WWW: An Overview, *ACM SIGGROUP Bulletin*, 21(3), pp. 3 - 12
- Mimos Berhad. (n.d.). *Agribazaar: Exchange for Better Price*. The Agribazaar, Retrieved Aug 13, 2007 from <u>http://www.agribazaar.com.my</u>
- Mimos Berhad. (n.d.). *Agribazaar: Exchange for Better Price*. The Agribazaar, Retrieved December 23, 2009 from <u>http://www.agribazaar.com.my</u>
- Myers, M.D. (1997). "Qualitative Research in Information Systems", MISQ Discovery. Retrieved August 16, 2007 from <u>http://www.misq.or/discovery/MISQDisworld/index.html</u>

Nor Iadah Yusop, Zahurin Mat Aji, Huda Ibrahim, Rafidah Abd. Razak, Wan Rozaini

Sheik Osman. (2007). Rural Internet Centre (RIC) as Catalysts for Building Knowledge-Based Society - The Case of Northern States of Malaysia. Proceedings from HCI International 2007, Beijing, China.

- Norliza Katuk, Nur Haryani Zakaria, Faudziah Ahmad, Wan Rozaini Sheik Osman, Nor Farzana Abd Ghani, Kamarul Faizal Hashim. (2007). "The Implementation of Web-based Application for Agribusiness in Malaysia".
 Proceeding in Rural ICT Development Conference (RICTD 2007), Kedah, Malaysia.
- Patton, M. Q. (2001). *Qualitative research and evaluation methods* (3rd ed.). Thousand Oaks, CA: Sage.
- Pearson, S.W. and Bailey, J.E. (1980). Measurement of computer user satisfaction, ACM SIGMETRICS Performance Evaluation Review, 9(1), pp. 59 - 68
- Punch, K.F. (2005). Introduction to Social Research–Quantitative & Qualitative Approaches. London: Sage.

Qualitative research n." A Dictionary of Nursing. Oxford University Press, 2008. Oxford Reference Online. Oxford University Press. Universiti Malaysia Perlis (UniMAP). 25 October 2008 <u>http://www.oxfordreference.com/views/ENTRY.html?subview=Main&</u> <u>entry=t62.e10220</u>

Saade, R and Bahli, B. (2005). The impact of cognitive absorption on perceived usefulness and perceived ease of use in on-line learning: an extension of the technology acceptance model. *Information & Management*, 42(2), pp. 317-327.

Schrock, K. (2006). Critical evaluation of a web site - Secondary school level. Retrieved July 7, 2006, from http://school.discovery.com/schrockguide/pdf/schrockwebevalsec.dot

- Schwab, D. P. and Cummings, L. L. (1973). Theories of Performance and Satisfaction: A Review, in *Readings in Organizational Behavior and Human Performance*, eds. W. E. Scott and L. L. Cummings. Homewood: Richard D. Irwin, Inc., 1973, pp. 130-53.
- Sekaran, U. (2000). *Research Method fo Business*. New York, NY: John Wiley & Sons Inc.
- Shklar, L. and <u>Rosen</u>, R. (2003). Web Application Architecture: Principles, Protocols and Practices, Wiley, ISBN 0-471-48656-6.
- Stanton, J. V. (2006). "The Role of Agribusiness in Development". Journal of Agribusiness. Vol. 18, No. 2, Spring 2000.
- Stone, G. (2005). Agribusiness Role in Extension, Education and Training : A Case Study. Research Report in Capacity Building in Rural Australia. Retrieved Nov 7, 2006 from <u>http://www.agribusiness.asn.au/index.php</u>
- Straub, D., Limayem, M. and Karahanna, E. (1995). "Measuring System Usage: Implications for IS Theory Testing," *Management Science*, vol. 41, pp. 1328-1342.
- Suruhanjaya Komunikasi dan Multimedia Malaysia. (2008). *Household use of the Internet Survey 2008*. Retrieved from <u>http://www.apira.org/data/upload/HUIS08_02_oXJL4a.pdf</u>
- Sutton, K. (2001). "Agribusiness On A Grand Scale Felda's Sahabat Complex in East Malaysia". Singapore Journal of Tropical Geography, Vol. 22, No. 1, pp. 90 - 105.
- U.S. Agribusiness Council, 2006, *Agribusiness Definition*. Retrieved Nov 7, 2006 from <u>http://www.agribusinesscouncil.org/#def</u>

Venkatesh, V., Morris, M.G., Davis, G.B. and Davis, F.B. (2003). "User

Acceptance of Information Technology: Toward a Unified View," *MIS Quarterly*, vol. 27, pp. 425-454.

- Venkatesh, V and Davis, F. (2000), 'A theoretical extension of the technology acceptance model :four longitudinal field studies', *management science*, 46 (2), pp.186-204.
- Venkatesh, V., et al. (2003), 'User acceptance of information technology: toward unified view', *MIS quarterly*, 27 (3), pp.425-78.
- Waksman, G., & Zanasi, C. (2003). Analysis Of Information Resources Dedicated To ICT On The Green Web In France And Italy. Paper presented at the EFITA 2003 Conference, Debrecen, Hungary.
- Wanous, J. P. and Lawler, E. E. (1972). Measurement and Meaning of Job Satisfaction, *Journal of Applied Psychology*, 56(2), pp. 95-105.
- Wixom, B. H., & Todd, P. A. (2005). A theoretical integration of user satisfaction and technology acceptance. Information Systems Research, 16(1), 85-102.
- Wong, L.C.Y. (2007). Development of Malaysia's Agricultural Sector: Agriculture as an Engine of Growth. ISEAS 'Conference on the Malaysian Economy: Development and Challenges' 25 -26 January, Singapore. Retrieved Sept 18, 2007 from http://www.isis.org.my/files/pubs/papers/Territorial_Disputes_in_East%20Asi a.pdf
- Ying, Z., Wang, Y. & Li, Z (2005). Research on circulation and E-Commerce of Chinese agricultural products in *Proceedings of ICEC'05*, August 15-17, 2005, Xi'an, China.
- Zahurin Mat Aji, Huda Ibrahim, Nor Iadah Yusop, Rafidah Abdul Razak, Wan Rozaini Sheik Osman. (2007). Rural Internet Centre (RIC) as a Mean for Bridging the Digital Gap in Rural Communities: Assessment of Current

Situations. In *Managing Worldwide Operations and Communications with Information Technology. IRMA 2007 Vancouver, Canada.* Khosrow-Pour, M (Ed). (pp. 1167-1171). Idea Group Publishing.

Zulkhairi Md. Dahalin, Suhaidi, Hassan, Huda Ibrahim, Wan Rozaini Sheik Osman, Azman Yassin, Md. Khairuddin Kasiran, Mohd Nizam Saad, Mohamad Amir Seman, Nor Iadah Md. Yusof, Osman Ghazalli, Abd. Razak Rahmat, and Suwannit Chareen Sop Chit. (2008). *Kajian Keperluan Aplikasi dan Kandung ICT (Penggunaan Jalur Lebar di Kawasan Luar Bandar Negeri Kedah and Perlis)*. Sintok: Universiti Utara Malaysia.

Appendix A: Screenshots of Agribazaar Portal

agriba Exchange For Bet	zaar er Price		
Laman Utama e-Juali	Beli e-Stok e-Logistik	e-Sokongan e-Pembuatan	
Inneinto Pendaftaran adalah PERCUMAI Daftarlah sekarang ID Pengguna Katalaluan Login Reset Anda Perlukan Bantuan? Jangkaan Pengeluaran Buah-Buahan Bermusim Jangkaan Helalui ATAU Jangkaan Philh buah V	Image: State	han Ikan Ovagreen 2010 Serta membekal hormon pembenihan/ aruhan ouatan t 2010 Minat untuk membeli ayam kampung hidup dan ama d 2010 organik yang dihasilkan dari tahi lembu tulen siap g Pernjagaan Kategori Tawaran Lagi Tawaran>>	Finalist 2006 Stockholm Challenge www.stockholmchallenges Pengumuan Hohen Maaf (2009-08-02) Pithek kami kamponia AgriBazar pada 29/709 hingga 2/8/09 yang lalu. Segala kesultan amat dikesali. email pengauna (2009-03-05) Pengguna yang mendafara bermula pada 1 Jun 2008 akam menghadapi masalah teknadap capalan email. menghadapi masalah login boleh menghubungi talan 03-88703175
Tawaran Popular Baru! 1. alatan fertigasi	BENIH TANAMAN(6)	BUAH-BUAHAN(70)	atau 03-88703108. Harap maklum. Masalah Login (2006-01-12) Sila hubungi terus ke talian 03-88703109, 03-88703175 atau 03-88703103
 projek usahasama pente, alatan tanaman cendawa, baja teknologi em bekalan lembu/kerbau d, bekalan hasil pertania, pelbaqai jenis kerepek, membeli semua jenis bu, membeli pelbaqai jenis, 	EKSTRAK MINUMAN KESIHATAN(4) bersa organik perang hydrolisis ,herba INPUT PERTANIAN(11) baja ,hormon PERKHIDMATAN(8) khidmat rundingan,pelancongan PRODUK MINUMAN(21) air moloumaa ius	HASUINIJaun Jamman HASU SENI DAN PERTUKANGAN(7) kraftangan,pertukangan kayu PERTIKANAN(88) ikan air masin,ikan air tawar PRODUK MAKANAN(160) hasilan laut,kerepek PRODUK TENUSU(8) penjangan kesihatan guju sener.	Statistik AgriBazaar Jumlah Ahli : 55822 Tawaran minggu ini : 5 Jumlah tawaran : 1010 <u>Kataloq</u> : 972 <u>Statistik kerperincl</u> <u>Statistik Lawatan</u>
Nama Q Perniagaan Q Tawaran : Q Q Negeri :	SAYUR-SAYURAN/TANAMAN LADANG(125) cendawan,sayuran jenis buah TANAMAN INDUSTRI(6) beverage,utama <u>TERNAKAN(27)</u> esotik,lain-ian	TANAH PERTANIAN(28) jusi/sewa TANAMAN PERHUTANAN(3) kayu Kayan	Harga Pasaran Ramalan Cuaca Tawaran Keria S Rahsia AgriBazaar AgriBazaar Facebook
Katalog Produk : Q Carian Terperinci Ruang Niaga • Papan Tawaran • Senarai Katalog	Industri herba wijid ramai usahawan baru (Memandangkan negara mempunyai khazana terkaya di dunia, sudah semestinya ia boleh i hasil untuk dimanfaatkan. 35,880 petani di seluruh Johor mena SERANAI 45,886 petani di seluruh Johor mena bayun jalu danga 6,423 metani terbahi dipat	2009-02-23) h kepelbagaian biologi flora dan fauna antara menjana kekayaan dalam konteks pengeluaran <u>ilan</u> (2008-07-01) daftar dengan jabatan pertanian negeri, sepanjang an 465 notek perdahanan caran berkahanan-i	Aktiviti Menarik Klinik pengembangan (Doktor Pokok) Di mana anda boleh mendapatkan khidmat Dr. Pokok ? Sila layari: <u>www.doa.gov.my</u>
Info AgriBazaar <u>Perihal AgriBazaar</u> <u>Bantuan</u> <u>Hubungi Kami</u> <u>Terma & Syarat</u>	Berbakti kegada tanah beri hasi lumayan (20 BATU PAHAT 28 April - Berbakti kepada tanah malah mampu memberi hasil lumayan. Berp menceburi bidang tersebut secara serius lebih sawit. Keledek Ganti Tembakau Di Pantai Timur 201 BACHOK: Kementerian Pertanian dan Industri negara mencapai hingga 300,000 tan setahur	 M. Song Hole Service Service (1998). M. Song Hole Service Se	Kalender Kursus Pertanian Jabatan Pertanian 2009 Pulau Pinang <u>Klik sini</u> Perlis Kedah Perak Kalantan <u>Klik sini</u>

Semua berita>> Kelantan<u>Klik sini</u>



<

Appendix B: Online Questionnaire

Part 1: Respondent's Background

Gender	: • Male ^C Female
Age	: Please choose your age ::
Marital Status	: Please choose your marital status ::
Race	: $^{\circ}$ Malay $^{\circ}$ Chinese $^{\circ}$ Indian $^{\circ}$ Others
Education Level	PMR SPM STPM Bachelor Degree Master Degree Others
Are you working?	: • Yes · No
Organization's Type	: Please choose the type of organization ::
Monthly Income	: Please choose your monthly income ::
Do you use Agribazaar?	: [○] Yes [●] No
How long have you used Agribazaar?	: Please choose the lenght of using Agribazaar ::
How did you know about Agribazaar?	: Please choose Agribazaar source of information ::
Do you register with Agribazaar?	: ^O Yes [®] No

Part 2: Respondent's Level of Acceptance

t 2: pondent's	The usin	following questions aim to determine your satisfa g Agribazaar portal.	actory level when					
eptance	Please choose the most appropriate rating. Your rating should range from 1 (strongly disagree) to 5 (strong agree) For items that are not applicable, use N/A							
	1.	The content in the website is comprehensive.	$\begin{array}{ccc} & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & &$					
	2.	Using the website increases my productivity.	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					
	3.	I will increase my chances of getting a business by using the website.	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					
	4.	The information is reliable and error-free.	$ \begin{bmatrix} \bigcirc & {}_1 \bigcirc & {}_2 \\ \bigcirc & {}_3 \bigcirc & {}_4 \\ \bigcirc & {}_5 \bigcirc & {}_{N/A} \end{bmatrix} $					
	5.	There is a way to contact the buyer/seller through the website.	$ \begin{bmatrix} \bigcirc & _{1} \bigcirc & _{2} \\ \bigcirc & _{3} \bigcirc & _{4} \\ \bigcirc & _{5} \bigcirc & _{N/A} \end{bmatrix} $					
	6.	Using the website is so complicated, it is difficult to understand what is going on.	$ \begin{bmatrix} \bigcirc & _1 \bigcirc & _2 \\ \bigcirc & _3 \bigcirc & _4 \\ \bigcirc & _5 \bigcirc & _{N/A} \end{bmatrix} $					
	7.	The information is clearly presented.	$ \begin{bmatrix} 0 & & & & \\ & 1 & & 2 \\ 0 & & 3 & & 4 \\ 0 & & 5 & & N/A \end{bmatrix} $					
	8.	There is a fee to obtain the information.	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					
	9.	You can move around the website easily and quickly.	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					
	10.	The website loads quickly.	$0_1 0_2$					

		0	5 °	N/A
11.	People who are important to me (family, friends & business partners) think that I should use the website.	0	1 O 3 O	2 4
		0	5 ^O	N/A
12.	In general, the government (Department of Agriculture, Malaysia) has supported the use of the website.	0	10	2
		0	3	4 N/A
13.	I am in a better position than those who do not		5	N/A
	use the website.	0	1 30	2 4
		0	50	' N/A
14.	My reputation/credibility with stakeholders has improved by using the website (i.e.	0	10	2
	relationship with government agency, banks,	0	3 O	4
	customers, retainers)	0	5 ^O	N/A
15.	I can interact with other buyers/sellers over the website at anytime.	0	1 O	2
		0	30	4
		0	5 ⁽⁾	N/A
16.	Given the resources, opportunities and knowledge it takes to use the website, it would	0	10	2
	be easy for the to use the website.	0	30	4 N/A
17.	The resources in the website have a reputable	0	э 1 ⁰	2
	expert benna it.	0	30	4
		0	50	N/A
18.	The website "grabs me" and makes me want to read more.	0	1 ^O	2
		0	3 O	4
		0	5 ^O	N/A
19.	Menu items are logically grouped.	0	10	2
			30	4
		U U	5 ⁽⁾	N/A
20.	Using the website is compatible with all aspects of my work.	0	10	2
			3	4
		\mathbb{P}	5 💛	N/A

	Your comments on Agribazaar: (Up to 50 characters)
	Your e-mail address:
Co	pyright © 2007 Universiti Utara Malaysia

Appendix C: SPSS Output

Variables Entered/Removed[®]

Model	Variables Entered	Variables Removed	Method
1	FCC, PRE, SOE, EFE	-	Enter

a. All requested variables entered.

b. Dependent Variable: USE

Model Summary

	_		Adjusted	Std. Error of
Model	R	R Square	R Square	the Estimate
1	.892 ^a	.796	.789	.32017

a. Predictors: (Constant), FCC, PRE, SOE, EFE

Coefficients^a

		Unstandardized Coefficients		Standardized Coefficients		
Model		В	St.d. Error	Beta	t	Sig.
1	(Constant)	.049	.201		.244	.807
	PRE	.770	.067	.214	2.888	.005
	EFE	.259	.082	.282	3.165	.002
	SOE	.239	.087	.242	2.744	.007
	FCC	.312	.067	.267	4.657	.000

a. Dependent Variable: USE

Correlations	Correl	lations
--------------	--------	---------

								Education		
		PRE	EFE	SOE	FCC	Gender	Age	Level	Income	USE
PRE	Pearson Correlation	1	.781**	.788**	.599**	010	037	.017	103	.786**
	Sig. (2-tailed)		.000	.000	.000	.917	.689	.858	.284	.000
	Ν	118	118	118	118	116	118	118	109	118
EFE	Pearson Correlation	.781**	1	.853**	.653**	.073	031	.040	023	.831**
	Sig. (2-tailed)	.000		.000	.000	.438	.737	.667	.810	.000
	Ν	118	118	118	118	116	118	118	109	118
SOE	Pearson Correlation	.788**	.853**	1	.620**	045	047	.053	.005	.818**
	Sig. (2-tailed)	.000	.000		.000	.628	.613	.571	.958	.000
	Ν	118	118	118	118	116	118	118	109	118
FCC	Pearson Correlation	.599**	.653**	.620**	1	.016	.035	.068	066	.730**
	Sig. (2-tailed)	.000	.000	.000		.867	.703	.466	.492	.000
	Ν	118	118	118	118	116	118	118	109	118
Gender	Pearson Correlation	010	.073	045	.016	1	.073	001	072	.041
	Sig. (2-tailed)	.917	.438	.628	.867		.435	.993	.459	.664
	Ν	116	116	116	116	117	117	117	108	116
Age	Pearson Correlation	037	031	047	.035	.073	1	018	.074	.018
	Sig. (2-tailed)	.689	.737	.613	.703	.435		.850	.442	.843
	Ν	118	118	118	118	117	119	119	110	118
Education Level	Pearson Correlation	.017	.040	.053	.068	001	018	1	.056	.048
	Sig. (2-tailed)	.858	.667	.571	.466	.993	.850		.561	.604
	Ν	118	118	118	118	117	119	119	110	118
Income	Pearson Correlation	103	023	.005	066	072	.074	.056	1	054
	Sig. (2-tailed)	.284	.810	.958	.492	.459	.442	.561		.579
	Ν	109	109	109	109	108	110	110	110	109
USE	Pearson Correlation	.786**	.831**	.818**	.730**	.041	.018	.048	054	1
	Sig. (2-tailed)	.000	.000	.000	.000	.664	.843	.604	.579	
	Ν	118	118	118	118	116	118	118	109	118

** Correlation is significant at the 0.01 level (2-tailed).

Appendix D: Interpretive Result

Comments from agribazaar users ID = 4.593 as a from fast neithing (f = 3)Total Comments = 590 Duplicate entries = 20 Irrelevant comments = 217 (most of the users used this platform to advertise products, post messages/announcement, responses-from agribazaar administrator) Total Entries Analyzed = 353 Total Comments Analyzed = 372. The total number of issues raised = 32

No	Issues	Reasons	Frequency
1	account blocked	incorrect password/login, certificate error, difficult/unable to log in (even though the password & log-in id is correct, sometimes the system asked users to re-register), tidak log-in selama sebulan, asked for help (i.e. to unblok) but did not get response, have been given user id but still could not log-in	REGISTER
2	website need to be regularly updated	date, inactive fax number, delete the garbage post, incomplete/wrong address, list of State Agriculture Dept, (tel. no, address, etc), links	
3	more promotion of the websites	better/more transactions (buying & selling), many people are unaware of the portal existence	- 13
4	services need to be upgraded	speed getting slower: browsing from page to page took longer time (10 minutes), after office hour service, should include more categories, filling up tender form should be easter, not properly maintained	8
5	online forum	for discussions on agriculture related matters	7

Issues highlighted from users' comments

1	terms & conditions should	1.215.5	_
21	ramban produk di bawah fajuk hesil buah unluk dijadikan tepung		
20	users information	e-stock, market price	
19	symbol for submenu	e-mail symbol is confusing	
18	portal can be more interesting if given support by successful businessmen/organizations		
17	portal should include more enimation (not static)		
16	more information should be included in the website	Incomplete information on various product (leach, palm, pokak misal kvoing, landscape plants etc.), product marketing, product packaging, pasticitie, lists of product in the market plice column is limited, makanan ikan, aquarium & aquatic plant, price of vegetables & fruits, agricultural techniques & technologies, horse breeding, hydroponic planting, mushroom planting, agriculture/farming into, more advertisement on products & training courses (ayam kampung).	
15	problems with users e-mail Gapribazaar.com.my	6-mail cannot be accessed, problem with snam	
14	products & training courses		
13	want to do business transaction (not for those who just want to browse the site> unregistered users)		2
12	display link to other pages on the current page		
1	1 make it more user friendly	difficulties in understanding certain procedures, difficulties in getting information	
1	unorganized categories of business directory	difficulties in registering type of business, list of livestock breaders should be categorized by states	
	9 website/porta/	provide a lot of help, useful for the farmers, useful directory	-
	acknowledgement of username & password after registration takes 3 8 days	Incine	
	6 registration problem 7 problem sending e-mail	some of the column cannot be filled, e-mail cannot be sent to the administrator	
		filling up the registration form (using a 3G phone), international candidate>do not have IC number, do not receive any confirmation, application rejected, tried 3 times but to no avail, after 3 months still to unsuccessful	

23	product descriptions should include with pictures		
	portal should include		
-24	English version	for international user & Mataysian Chinese	6
25	training should be given to the farmers		3
28	add another function that will enable the users to upload pictures of their products		4
37	should enable linkages to other agriculture agencies, international traders		
	the system does not provide ways to recall		
28	password dease include section for		2
29	recipes		1
30	if possible to include hardware trading		
31	market price page cannot be opened		2
32	should have a full time agriculture officer to response to users' queries		1
33	seharusnya semua iklan jualan/belian yang lalu, yang masih sah Tawarannya, diklan secara berterusan melalui rolling screen		
34	no/slow action/response from MIMOS/admin regarding comments/feedbacks from users		7
-	Include a section of		1
35	general trading		1
37	problem with tender/offer (tawaran)/advertisement	cannot be accessed/opened/done (problem with calendar), editing cannot be done, cannot activate the tender form that has been filled, index is not included >do no know the number of people hit the offer, price of the product need to be stated.	42
38	more categories should be included in the tender/offer section.	unable to find the herbs category cocces. fishball, buil frog	5
39	bahagian pertanyaan pada penawar tidak boleh digunakan		1
40	agriculture dept. at all states should provide access to agribazeer for		1

	farmers		
	response from users are	lots of manufacturers & farmers do not participate	1
41	very slow		- 4
42	how to list product in the catalog		1
10	change the format in the sell/buy page so that more		1
43	names can be displayed		4
44	security	search, the product management button is not	3
45	catalog	functioning, should be in a partorni	4
46	searching functions	not working, could not find "serai wangi TOTAL	372